Scientific Quality: Grade C (Good)
Novelty of This Manuscript: Grade C (Fair)
Creativity or Innovation of This Manuscript: Grade B (Good)
Scientific Significance of the Conclusion in This Manuscript: Grade B (Good)
Language Quality: Grade B (Minor language polishing)

Specific Comments to Authors: Labor analgesia technology is mainly the use of epidural anesthesia for analgesia. However, anesthetics block the sacral plexus. In this study, transcutaneous electrical stimulation of Baliao points combined with biofeedback therapy was used to intervene in parturients with labor analgesia. In this study, the authors explored the application effect of transcutaneous electrical stimulation at Baliao point combined with biofeedback in preventing postpartum urinary retention in parturients undergoing labor analgesia. The study is well performed, and the results are interesting. However, some minor revisions are required before acceptance: 1. The manuscript requires a minor editing. Some spelling mistakes should be corrected. 2. The title is too long, please short it. 3. Please move the limit of the study to the end of the discussion from the conclusion. 4. Figure 1 should be improved.

Reply:
1. Spelling mistakes in the text have been corrected.
2. The title has been shortened.
3. We have moved the limitations of our research from the conclusion to the end of the discussion.
4. We have improved Figure 1 and provided a PPT.